Author's response to reviews

Title: The effect of scheduled antibody testing on treatment patterns in interferon-treated patients with multiple sclerosis

Authors:

Edward Fox (FoxTexMS@gmail.com)
Barbara Green (bjgreenmd@aol.com)
Clyde Markowitz (cmarkowi@mail.med.upenn.edu)
Ronald Murray (murrayneuro@gmail.com)
Andrew Goodman (andrew_goodman@urmc.rochester.edu)
Stephen Glenski (steve.glenski@tevapharm.com)
Pippa Loupe (Pippa.Loupe@tevapharm.com)
J. Nita Cogburn (nita.cogburn@tevapharm.com)

Version: 2 Date: 19 November 2013

Author's response to reviews:

Dear Dr. Josefino M. Rodis,

We have updated the attached manuscript per the editorial formatting requirements listed below.

Kindest Regards, Dr. Fox

1. Please note that Research article require the following sections: Conclusions. Please ensure that these sections are present and clearly labelled as described above. Please do check the instructions for authors on the journal website to ensure that your manuscript follows the correct structure for this journal and article type, and to ensure that you are aware of additional recommendations for formatting that will facilitate handling of your manuscript.